UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-K/A-1

☑ ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the fiscal year ended December 31, 2015

OR ☐ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 For the transition period from_____ to ___ Commission file number 1-12830

BioTime, Inc. (Exact name of registrant as specified in its charter)

California 94-3127919

(State or other jurisdiction of incorporation or organization)

(I.R.S. Employer Identification No.)

1010 Atlantic Avenue, Suite 102 Alameda, California 94501

(Address of principal executive offices) (Zip Code)

Registrant's telephone number, including area code (510) 521-3390

Securities registered pursuant to Section 12(b) of the Act

Title of each class

Common shares, no par value Common share purchase warrants expiring October 1, 2018 Name of exchange on which registered

NYSE MKT NYSE MKT

Securities registered pursuant to Section 12(g) of the Act:

None

Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes \square No \boxtimes

	Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes \square No \boxtimes
1934 du	Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of tring the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing nents for the past 90 days. Yes \boxtimes No \square

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data Fil required to be submitted and posted pursuant to Rule 405 of Regulation S-T (\S 232.405 of this chapter) during the preceding 12 months (or for such shorted period that the registrant was required to submit and post such files). Yes \square No \square		
Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 and will not be contained, to the best of registrant's knowledge, in definitive proxy or Form 10-K or any amendment to this Form 10-K ⊠		
Indicate by check mark whether the registrant is a large accelerated filer, an acc See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting		
Large accelerated filer □	Accelerated filer ⊠	
Non-accelerated filer (Do not check if a smaller reporting company)	Smaller reporting company □	
Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act): Yes \square No \boxtimes The approximate aggregate market value of voting common shares held by non-affiliates computed by reference to the price at which common shares were last sold as of June 30, 2015 was \$165,569,534. Shares held by each executive officer and director and by each person who beneficially own more than 5% of the outstanding common shares have been excluded in that such persons may under certain circumstances be deemed to be affiliates. This determination of affiliate status is not necessarily a conclusive determination for other purposes.		
The number of common shares outstanding as of I	March 4, 2016 was 94,894,140.	
Documents Incorporated by	Reference	
Portions of the registrant's Proxy Statement for 2016 Annual Meeting of	Shareholders are incorporated by reference in Part III	

EXPLANATORY NOTE

BioTime, Inc. ("BioTime") is filing this Amendment No. 1 (the "Amendment") to its Annual Report on Form 10-K for the period ended December 31, 2015, as originally filed with the Securities and Exchange Commission on March 15, 2016 (the "Original Filing"), for the sole purpose of filing a revised redacted version of Exhibit 10.68, reflecting changes to BioTime's confidential treatment request with respect to certain portions of the exhibit. No other changes have been made to the Original Filing or any other exhibits.

This Amendment does not reflect events occurring after the date of the Original Filing or modify or update any disclosures that may have been affected by subsequent events.

Item 15. Exhibits, Financial Statement Schedules

(a-1) Financial Statements.

The following financial statements of BioTime, Inc. are filed in the Form 10-K:

Consolidated statements of operations Consolidated statements of shareholders' deficit Consolidated statements of cash flows

Notes to Financial Statements

(a-2) Financial Statement Schedules

All schedules are omitted because the required information is inapplicable or the information is presented in the financial statements or the notes thereto.

(a-3) Exhibits.

Exhibit

Numbers	<u>Description</u>
2.1	Asset Contribution Agreement, dated January 4, 2013, by and among BioTime, Inc., BioTime Acquisition Corporation, and Geron Corporation. Schedules to the Asset Contribution Agreement have been omitted. BioTime agrees to furnish supplementally a copy of the omitted schedules to
	the Commission upon request. (1)
2 1	Articles of Learn antique with all amondments (2)
3.1	Articles of Incorporation with all amendments (2)
3.2	By-Laws, As Amended (3)
4.1	Specimen of Common Share Certificate (4)
4.2	Form of Warrant Issued June 2013 (5)
4.2	Warrant American Lated as of October 1 2012, as a mondad Contamber 10, 2014, between DisTimes Inc. and American Steels Transfer 6, Transfer
4.3	Warrant Agreement, dated as of October 1, 2013, as amended September 19, 2014, between BioTime, Inc. and American Stock Transfer & Trust Company, LLC as Warrant Agent for the benefit of Asterias Biotherapeutics, Inc. (6)
	Company, LLC as warrant Agent for the benefit of Asterias Biotherapeuties, file. (6)
4.4	Warrant Issued October 1, 2013 to Asterias Biotherapeutics, Inc. (included in Exhibit 4.7) (6)
10.1	Intellectual Property Agreement between BioTime, Inc. and Hal Sternberg (4)
10.0	
10.2	Intellectual Property Agreement between BioTime, Inc. and Judith Segall (4)
10.3	2002 Stock Option Plan, as amended (7)
10.5	2002 Stock Option 1 ran, as amended (7)
	3

	for confidential treatment) (8)
10.5	Modification of Exclusive License Agreement between Abbott Laboratories and BioTime, Inc. (Portions of this exhibit have been omitted pursuant to a request for confidential treatment) (9)
10.6	Exclusive License Agreement between BioTime, Inc. and CJ Corp. (10)
10.7	Amendment to Exclusive License Agreement Between BioTime, Inc. and Hospira, Inc. (11)
10.8	Employment Agreement, dated October 10, 2007, between BioTime, Inc. and Michael D. West. (12)
10.9	Commercial License and Option Agreement between BioTime and Wisconsin Alumni Research Foundation (13)
10.10	License Agreement, dated July 10, 2008, between Embryome Sciences, Inc. and Advanced Cell Technology, Inc. (14)
10.11	First Amendment of Commercial License and Option Agreement, dated March 11, 2009, between BioTime and Wisconsin Alumni Research Foundation (15)
10.12	Registration Rights Agreement between OncoCyte Corporation and George Karfunkel (16)
10.13	Amended and Restated Shareholders Agreement, dated October 7, 2010, by and among ES Cell International Pte. Ltd, BioTime, Inc., Teva Pharmaceutical Industries, Limited, HBL-Hadasit Bio-Holdings, Ltd., and Cell Cure Neurosciences Ltd. (17)
10.14	Amended and Restated Research and License Agreement, dated October 7, 2010, between Hadasit Medical Research Services and Development Ltd. and Cell Cure Neurosciences Ltd. (17)
10.15	Additional Research Agreement, dated October 7, 2010, between Hadasit Medical Research Services and Development Ltd. and Cell Cure Neurosciences Ltd. (17)
10.16	Exclusive License Agreement, dated November 20, 2007, between Cell Targeting, Inc. and Burnham Institute for Medical Research (17)
10.17	OncoCyte Corporation 2010 Stock Option Plan; Form of OncoCyte Corporation Stock Option Agreement (17)
10.18	OrthoCyte Corporation 2010 Stock Option Plan; Form of OrthoCyte Corporation Stock Option Agreement (17)
10.19	BioTime Asia, Limited 2010 Stock Option Plan; Form of BioTime Asia Limited Stock Option Agreement (17)
	4

Exclusive License Agreement between Abbott Laboratories and BioTime, Inc. (Portions of this exhibit have been omitted pursuant to a request

10.4

10.20	License Agreement between BioTime, Inc. and Cornell University (Portions of this exhibit have been omitted pursuant to a request for confidential treatment) (18)
10.21	LifeMap, Inc. 2011 Stock Option Plan; and Form of LifeMap, Inc. Stock Option Agreement (19)
10.22	Exclusive License Agreement, dated February 15, 2006, between Glycosan BioSystems, Inc. and the University of Utah Research Foundation, as amended (20)
10.23	Option Agreement, dated June 3, 2013, between BioTime, Inc. and certain investors (21)
10.24	Client Referral and Solicitation Agreement, dated April 1, 2013, between BioTime, Inc., LifeMap Sciences, Inc. and OBEX Securities, LLC (5)
10.25	Royalty Agreement, dated October 1, 2013, between Asterias Biotherapeutics, Inc. and Geron Corporation (22)
10.26	Exclusive Sublicense Agreement, dated October 1, 2013, between Geron Corporation and Asterias Biotherapeutics, Inc. (22)
10.27	Exclusive License Agreement, dated February 20, 2003, and First Amendment thereto dated September 7, 2004, between The Regents of the University of California and Geron Corporation (22)
10.28	Non-Exclusive License Agreement, dated as of October 7, 2013, between the Wisconsin Alumni Research Foundation and Asterias Biotherapeutics, Inc. (Portions of this exhibit have been omitted pursuant to a request for confidential treatment) (22)
10.29	Equity Incentive Plan (22)
10.30	Form of Employee Incentive Stock Option Agreement (22)
10.31	Form of Non-employee Director Stock Option Agreement (22)
10.32	Asterias Biotherapeutics, Inc. Equity Incentive Plan (23)
10.33	Form of Asterias Biotherapeutics, Inc. Employee Incentive Stock Option Agreement (24)
10.34	Form of Asterias Biotherapeutics, Inc. Non-employee Director Stock Option Agreement (24)
10.35	Lease, dated December 30, 2013, by and between BMR 6300 Dumbarton Circle, LP, and Asterias Biotherapeutics, Inc. (25)

10.36	Option Agreement, dated March 4, 2014, between BioTime and certain investors (25)
10.37	Co-Development and Option Agreement, dated May 6, 2014, between LifeMap Solutions, Inc. and the Icahn School of Medicine at Mount Sinai (Portions of this exhibit have been omitted pursuant to a request for confidential treatment) (26)
10.38	Stock Purchase Agreement, dated May 6, 2014, between LifeMap Sciences, Inc. and BioTime, Inc. (Portions of this exhibit have been omitted pursuant to a request for confidential treatment) (26)
10.39	Stock Purchase Agreement, dated June 12, 2014, between Pedro Lichtinger and Asterias Biotherapeutics, Inc. (26)
10.40	Purchase Agreement, dated June 13, 2014, between Broadwood Partners, L.P. and Asterias Biotherapeutics, Inc. (26)
10.41	Purchase Agreement, dated June 13, 2014, between The George Karfunkel 2007 Grantor Trust #1 and Asterias Biotherapeutics, Inc. (26)
10.42	Registration Rights Agreement, dated June 16, 2014, between The George Karfunkel 2007 Grantor Trust #1, Broadwood Partners, L.P., and Asterias Biotherapeutics, Inc. (26)
10.43	Employment Agreement, dated as of June 9, 2014, between Pedro Lichtinger and Asterias Biotherapeutics, Inc. (26)
10.44	LifeMap Solutions, Inc. 2014 Stock Option Plan (26)
10.45	Form of LifeMap Solutions, Inc. Incentive Stock Option Agreement (26)
10.45 10.46	Form of LifeMap Solutions, Inc. Incentive Stock Option Agreement (26) Form of LifeMap Solutions, Inc. Stock Option Agreement (26)
10.46	Form of LifeMap Solutions, Inc. Stock Option Agreement (26) Clinical Trial and Option Agreement, dated September 8, 2014, between Asterias Biotherapeutics, Inc. and Cancer Research UK and Cancer
10.46	Form of LifeMap Solutions, Inc. Stock Option Agreement (26) Clinical Trial and Option Agreement, dated September 8, 2014, between Asterias Biotherapeutics, Inc. and Cancer Research UK and Cancer Research Technology Limited(Portions of this exhibit have been omitted pursuant to a request for confidential treatment) (27) Notice of Grant Award, dated as of October 16, 2014, between the California Institute for Regenerative Medicine and Asterias Biotherapeutics,
10.46 10.47 10.48	Form of LifeMap Solutions, Inc. Stock Option Agreement (26) Clinical Trial and Option Agreement, dated September 8, 2014, between Asterias Biotherapeutics, Inc. and Cancer Research UK and Cancer Research Technology Limited(Portions of this exhibit have been omitted pursuant to a request for confidential treatment) (27) Notice of Grant Award, dated as of October 16, 2014, between the California Institute for Regenerative Medicine and Asterias Biotherapeutics, Inc. (Portions of this exhibit have been omitted pursuant to a request for confidential treatment) (28) Amendment to Notice of Grant Award, dated as of November 26, 2014, between the California Institute for Regenerative Medicine and Asterias

10.51	Employment Agreement, dated December 29, 2014, between BioTime, Inc. Aditya Mohanty (28)
10.52	Subscription Agreements between Asterias Biotherapeutics, Inc. and the investors named therein (28)
10.53	First Amendment to Co-Development and Option Agreement, dated March 7, 2015, between Icahn School of Medicine at Mount Sinai and LifeMap Solutions, Inc. (Portions of this exhibit have been omitted pursuant to a request for confidential treatment) (29)
10.54	2012 Equity Incentive Plan, as amended (30)
10.55	Stock Purchase Agreements, dated September 14, 2015, between BioTime, Inc. and certain investors (31)
10.56	Research & Development Agreement, dated September 29, 2015, between OrthoCyte Corporation and Heraeus Medical GmbH (Portions of this exhibit have been omitted pursuant to a request for confidential treatment) (31)
10.57	License Agreement, dated September 29, 2015, between OrthoCyte Corporation and Heraeus Medical GmbH (Portions of this exhibit have been omitted pursuant to a request for confidential treatment) (31)
10.58	Stock Purchase Agreements Between BioTime and certain investors (31)
10.59	Subscription Agreement, dated September 29, 2015, between OncoCyte Corporation and BioTime, Inc. (31)
10.60	Letter Agreement, dated September 24, 2015, between BioTime, Inc. and Union Underwriting & Finances Ltd. (32)
10.61	Employment Agreement, dated November 16, 2015, between BioTime, Inc. and Russell Skibsted (33)
10.62	Employment Termination and Release Agreement, dated November 18, 2015, between BioTime, Inc. and Robert W. Peabody (34)
10.63	Employment Agreement, dated November 18, 2015, between LifeMap Solutions, Inc. and Robert W. Peabody (34)
10.64	Consulting Agreement, dated November 18, 2015, between BioTime, Inc. and Robert W. Peabody (34)
10.65	Amendment of Employment Agreement, dated November 24, 2015, between BioTime, Inc. and Michael D. West (35)

10.66	Amendment of Employment Agreement, dated November 24, 2015, between BioTime, Inc. and Aditya Mohanty (35)
10.67	Lease, dated December 10, 2015, between BioTime, Inc. and BSREP Marina Village Owner LLC (36)
10.68	License Agreement, dated January 22, 2016, between OncoCyte Corporation and The Wistar Institute of Anatomy and Biology (Portions of this exhibit have been omitted pursuant to a request for confidential treatment) *
10.69	First Amendment to License Agreement, dated January 25, 2016, between OncoCyte Corporation and The Wistar Institute of Anatomy and Biology (37)
21.1	List of Subsidiaries (37)
23.1	Consent of OUM & Co. LLP (37)
23.2	Consent of Rothstein Kass (37)
<u>31</u>	Rule 13a-14(a)/15d-14(a) Certification *
32	Section 1350 Certification (37)
101	Interactive Data File
101.INS	XBRL Instance Document (37)
101.SCH	XBRL Taxonomy Extension Schema (37)
101.CAL	XBRL Taxonomy Extension Calculation Linkbase (37)
101.LAB	XBRL Taxonomy Extension Label Linkbase(37)
101.PRE	XBRL Taxonomy Extension Presentation Linkbase(37)
101.DEF	XBRL Taxonomy Extension Definition Document(37)
(1)	Incorporated by reference to BioTime's Current Report on Form 8-K filed with the Securities and Exchange Commission on January 8, 2013

(2) Incorporated by reference to BioTime's Annual Report on Form 10-K/A-1 for the year ended December 31, 2013 (3) Incorporated by reference to Registration Statement on Form S-1, File Number 33-48717 and Post-Effective Amendment No. 1 thereto filed with the Securities and Exchange Commission on June 22, 1992, and August 27, 1992, respectively Incorporated by reference to Registration Statement on Form S-1, File Number 33-44549 filed with the Securities and Exchange Commission on (4) December 18, 1991, and Amendment No. 1 and Amendment No. 2 thereto filed with the Securities and Exchange Commission on February 6, 1992 and March 7, 1992, respectively (5) Incorporated by reference to BioTime's Current Report on Form 8-K filed with the Securities and Exchange Commission on June 3, 2013 (6) Incorporated by reference to BioTime's Current Report on Form 8-K filed with the Securities and Exchange Commission on September 23, 2014 (7) Incorporated by reference to BioTime's Quarterly Report on Form 10-Q for the quarter ended June 30, 2009 Incorporated by reference to BioTime's Current Report on Form 8-K, filed April 24, 1997 (8)(9) Incorporated by reference to BioTime's Form Quarterly Report on 10-Q for the quarter ended June 30, 1999 (10)Incorporated by reference to BioTime's Annual Report on Form 10-K/A-1 for the year ended December 31, 2002 (11)Incorporated by reference to BioTime's Current Report on Form 8-K, filed January 13, 2006 (12)Incorporated by reference to BioTime's Annual Report on Form 10-KSB for the year ended December 31, 2007 (13)Incorporated by reference to BioTime's Current Report on Form 8-K, filed January 9, 2008 (14)Incorporated by reference to BioTime's Quarterly Report on Form 10-Q for the quarter ended June 30, 2008 (15)Incorporated by reference to BioTime's Annual Report on Form 10-K for the year ended December 31, 2008 Incorporated by reference to BioTime's Quarterly Report on Form 10-Q for the quarter ended September 30, 2009 (16)

(17)	Incorporated by reference to BioTime's Annual Report on Form 10-K for the year ended December 31, 2010
(18)	Incorporated by reference to BioTime's Quarterly Report on Form 10-Q for the quarter ended September 30, 2011
(19)	Incorporated by reference to BioTime's Annual Report on Form 10-K for the year ended December 31, 2011
(20)	Incorporated by reference to BioTime's Quarterly Report on Form 10-Q for the quarter ended September 30, 2012
, ,	
(21)	Incorporated by reference to BioTime's Quarterly Report on Form 10-Q for the quarter ended June 30, 2013
(22)	Incorporated by reference to BioTime's Quarterly Report on Form 10-Q for the quarter ended September 30, 2013
(23)	Incorporated by reference to Amendment No. 1 to Registration Statement on Form S-1 (333-187706) filed by Asterias Biotherapeutics, Inc. with the Securities and Exchange Commission on June 26, 2013
(2.1)	
(24)	Incorporated by reference to Amendment No. 2 to Registration Statement on Form S-1 (333-187706) filed by Asterias Biotherapeutics, Inc. with the Securities and Exchange Commission on August 13, 2013
(25)	Incorporated by reference to BioTime's Annual Report on Form 10-K for the year ended December 31, 2013
(26)	Incorporated by reference to BioTime's Quarterly Report on Form 10-Q for the quarter ended June 30, 2014
(20)	morporation by resource to 220 rand by Quanton, respect on 10 and 10 Quanton change on 201.
(27)	Incorporated by reference to BioTime's Quarterly Report on Form 10-Q/A-1 for the quarter ended September 30, 2014
(28)	Incorporated by reference to BioTime's Annual Report on Form 10-K for the year ended December 31, 2014
(26)	incorporated by reference to BioTime's Annual Report on Portin 10-K for the year ended December 51, 2014
(29)	Incorporated by reference to BioTime's Quarterly Report on Form 10/Q for the quarter ended March 31, 2015
(20)	Incomposited by reference to Designation Statement on Form S. 1. File Number 222 205661 filed with the Securities and Evahones Commission
(30)	Incorporated by reference to Registration Statement on Form S-1, File Number 333-205661 filed with the Securities and Exchange Commission on July 15, 2015
(2.1)	
(31)	Incorporated by reference to BioTime's Quarterly Report on Form 10/Q for the quarter ended September 30, 2015
	10

(32)	Incorporated by reference to Current Report on Form 8-K filed with the Securities and Exchange Commission on September 25, 2015
(22)	
(33)	Incorporated by reference to BioTime's Current Report on Form 8-K, filed November 16, 2015
(34)	Incorporated by reference to BioTime's Current Report on Form 8-K, filed November 18, 2015
(35)	Incorporated by reference to BioTime's Current Report on Form 8-K, filed November 24, 2015
(36)	Incorporated by reference to BioTime's Current Report on Form 8-K, filed December 9, 2015
(37)	Previously filed with BioTime's Annual Report on Form 10-K for the year ended December 31, 2015, filed March 15, 2016
*	Filed herewith
	riica neiewitii

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the Registrant has duly caused this Amendment No. 1 on Form 10-K/A to be signed on its behalf by the undersigned, thereunto duly authorized on the 24th day of May 2016.

BIOTIME, INC.

By: /s/Michael D. West

Michael D. West, Ph.D. Co-Chief Executive Officer

By: /s/Aditya Mohanty

Aditya Mohanty

Co-Chief Executive Officer

LICENSE AGREEMENT

This LICENSE AGREEMENT (the "<u>Agreement</u>") is made as of the 22nd day of January, 2016 (the "<u>Effective Date</u>"), by and between THE WISTAR INSTITUTE OF ANATOMY AND BIOLOGY, a nonprofit corporation organized and existing under the laws of the Commonwealth of Pennsylvania located at 3601 Spruce Street, Philadelphia, PA 19104 ("<u>Wistar</u>"), and ONCOCYTE CORPORATION, a corporation organized and existing under the laws of the State of California, with a principal place of business located at 1301 Harbor Bay Parkway, Alameda, CA 94502 ("<u>Company</u>").

BACKGROUND

WHEREAS, Wistar wishes to license to Company certain patent rights and technical information and know-how relating to molecular diagnostics for lung cancer, including, but not limited to confirmatory, companion and recurrence diagnostics for any type of lung cancer with detection through whole blood, fractionated blood, plasma, serum and/or other biological samples;

WHEREAS, Company wishes to obtain a license under such patent rights, technical information and know-how to develop and commercialize products, and Company is capable of and committed to developing and commercializing products utilizing such rights; and

WHEREAS, Wistar is willing to grant such a license to Company, in consideration of Company's satisfaction of its obligations hereunder, and for other good and valuable consideration as set forth below.

NOW, THEREFORE, in consideration of the premises and mutual covenants contained herein, and intending to be legally bound hereby, the parties hereto agree as follows:

ARTICLE 1 - DEFINITIONS

The following terms, as used herein, shall have the following meanings:

- 1.1 "Affiliate(s)" means, when used with reference to Company, any Person directly or indirectly controlling, controlled by or under common control with, Company. For the purposes of this definition, "control" means the direct or indirect ownership of over fifty percent (50%) of the outstanding voting securities of a Person, or the right to receive over fifty percent (50%) of the profits or earnings of a Person, or the ability to control the decisions of a Person.
 - 1.2 "Agreement" has the meaning set forth in the Preamble.

Exclusive License Agreement Wistar/OncoCyte

- "Bankruptcy Event" means, with respect to any Person, any of the following: such Person shall commence a voluntary case or other proceeding seeking liquidation, reorganization or other relief with respect to itself or its debts under any bankruptcy, insolvency or other similar law now or hereafter in effect or seeking the appointment of a trustee, receiver, liquidator, custodian or other similar official of it or any substantial part of its property, or shall consent to any such relief or to the appointment of or taking possession by any such official in an involuntary case or other proceeding commenced against it, or shall make a general assignment for the benefit of creditors, or shall fail generally to pay its debts as they become due, or shall take any corporate action to authorize any of the foregoing;
- an involuntary case or other proceeding shall be commenced against such Person seeking liquidation, reorganization or other relief with respect to it or its debts under any bankruptcy, insolvency or other similar law now or hereafter in effect or seeking the appointment of a trustee, receiver, liquidator, custodian or other similar official of it or any substantial part of its property, and such involuntary case or other proceeding shall remain undismissed and unstayed for a period of sixty (60) days; or an order for relief shall be entered against such Person under the federal bankruptcy laws as now or hereafter in effect; or
 - 1.3.3 a receiver or trustee shall be appointed with respect to such Person or all or substantially all of the assets of such Person.
 - 1.4 "Bar Date" has the meaning set forth in Section 7.1.1.
- "Calendar Quarter" means each three (3) month period, or any portion thereof, beginning on January 1, April 1, July 1 and October 1 of 1.5 each year.
 - 1.6 "Change of Control" has been intentionally omitted and is not referenced in this Agreement.
- 1.7 "Clinical Trials" means any administration of Licensed Products to humans for the purpose of demonstrating the safety or efficacy of the Licensed Product.
 - 1.8 "Commercialization Plan" has the meaning set forth in Section 4.2.1.
 - "Company" has the meaning set forth in the Preamble. 1.9
 - 1.10 "Company Confidential Information" has the meaning set forth in Section 8.1.2.
 - 1.11 "Company IP" has the meaning set forth in Section 9.6.3.
 - 1.12 "Confidential Information" has the meaning set forth in Section 8.1.3.
- 1.13 "Control" or "Controlled" means, with respect to the Technical Information, possession of the ability (whether by sole, joint or other ownership interest, license or otherwise, other than pursuant to this Agreement) to, without violating the terms of any agreement with a third party, grant a license or sublicense or provide access or other rights in, to or under such Technical Information.

Exclusive License Agreement Wistar/OncoCyte

Wistar Reference No. LIC15-35

1.3

- 1.14 "Courts" has the meaning set forth in Section 10.9.3.
- 1.15 "<u>Documentation and Approvals</u>" has the meaning set forth in Section 9.6.1.
- 1.16 "Effective Date" has the meaning set forth in the Preamble.
- 1.17 "First Commercial Sale" means the first sale, transfer, disposition, performance, or practice for value of a Licensed Product.
- 1.18 "Funding Agencies Interest" has the meaning set forth in Section 2.2.2.
- 1.19 "IDE Filing" means the preparation and filing of an "Investigational Device Exemption (IDE)", Premarket Notification 510(k), Premarket Approval, or similar marketing application or any comparable filing in a foreign jurisdiction.
 - 1.20 "Indemnified Party" and "Indemnified Parties" have the meaning set forth in Section 6.3.
 - 1.21 "<u>Liability</u>" and "<u>Liabilities</u>" have the meaning set forth in Section 6.3.
- 1.22 "<u>Licensed Field</u>" means molecular diagnostics for lung cancer, including, but not limited to confirmatory, companion and recurrence diagnostics for any type of lung cancer with detection through whole blood, fractionated blood, plasma, serum and/or other biological samples.
- 1.23 "<u>Licensed Patent(s)</u>" means (i) the U.S. patent applications set forth on Exhibit A attached hereto, (ii) all patents issuing from such applications, (iii) all continuations, continuations in part (to the extent that the claims are directed to inventions which are fully supported by the patents and patent applications of (i), (ii) and (iii)), additions, divisions, renewals, extensions, reexaminations and reissues that claim the benefit of the applications or patents referenced in (i) or (ii) hereof, and (iv) all foreign counterparts of any of the foregoing.
 - 1.24 "<u>Licensed Product(s)</u>" means:
- 1.24.1 on a country-by-country basis, any product, the making, using, selling, offering for sale, or importing of which product in the country in question would (without the license granted under this Agreement) infringe at least one pending Valid Claim (were it to have issued) or issued Valid Claim of the Licensed Patents in that country;
- 1.24.2 on a country-by-country basis, any service, process or method, the performing or providing of which process or method in the country in question would (without the license granted under the Agreement) infringe at least one pending Valid Claim (were it to have issued) or issued Valid Claim of the Licensed Patents in that country; and

Exclusive License Agreement Wistar/OncoCyte

- any product or process that is not covered by the foregoing clauses (i) or (ii), but that incorporates or is made, identified, developed, optimized, characterized, selected, derived or determined to have utility, in whole or in part, by the use or modification of (a) any Licensed Patent or any technology or invention covered thereby, (b) any Technical Information or (d) any Licensed Product covered by the foregoing clauses (i) or (ii).
- "Net Sales" means for each Licensed Product for any period, the gross amount billed or invoiced by Company, its Affiliates and Sublicensees less the following deductions: (i) customary trade, quantity and cash discounts actually allowed for the Licensed Product, (ii) rebates or retroactive price reductions actually allowed for the Licensed Product, or chargebacks actually made for the Licensed Product to federal, state or local governments (or their agencies) or any third party payor, administrator or contractor, including managed health organizations, or any other bona fide chargebacks actually made by Company for the Licensed Product that effectively reduces the selling price; (iii) taxes levied on sale or transportation of the Licensed Product and paid by or on behalf of Company, its Affiliates, or Sublicensees, (iv) freight allowances, insurance and custom duties for the Licensed Product, to the extent any of the foregoing subsections (i), (iii) and (iv) are separately stated on invoices for the Licensed Product, and (v) amounts that have been billed or invoiced but are written off as uncollectible (determined in a manner consistent with generally accepted accounting principles, consistently applied) due to insurance coverage limits, contracted in-network claims, denied claims, user co-pay and deductible fees and patient assistance fees, any such deduction to be taken in the royalty reporting period in which such amounts are written off, provided that should any such amounts thereafter be received, such amounts shall be recorded as Net Sales in the royalty reporting period of receipt. In the case of a sale or other transfer of a Licensed Product for which Company or a Sublicensee does not bill, Net Sales shall mean the amount received by Company and Sublicensees for the sale of such Licensed Product. If a Licensed Product is sold or otherwise transferred for consideration other than solely cash (whether or not at a discount), or if Licensed Product is billed or otherwise sold at a discounted price that is substantially lower than the customary prices charged by Company or Sublicensee, Net Sales shall be calculated based on the average amount charged for such Licensed Product in an arms-length transaction to a non-Affiliate, independent third party during the same Calendar Quarter in the same country or, in the absence of such sales, on the fair market value (as defined in Section 1.26 below) of such Licensed Product.
- 1.26 "Non-Royalty Sublicensing Income" means the fair market value of any and all consideration received by Company and its Affiliates from Sublicensees (or which Company is entitled to receive, whether or not offset against amounts payable to Sublicensee under the Sublicense) under or otherwise in connection with its Sublicenses, including license issue fees, lump sum payments and other licensing fees, option fees, milestone payments, minimum annual royalties, distribution fees, joint marketing fees, equity or other payments of any kind whatsoever, irrespective of whether such consideration is received in the form of cash, barter, credit, stock, warrants, release from debt, goods or services, licenses back, a premium on the sale of equity (i.e., payments for equity that exceed the pre-Sublicense fair market value of the equity), equity exchanges, or any other form whatsoever. Notwithstanding the foregoing, Non-Royalty Sublicensing Income specifically excludes the following: (i) royalties on Net Sales pursuant to Section 3.2; (ii) payments made by Sublicensee as consideration for the issuance of equity or debt securities of Company at the pre-Sublicense fair market value, provided that if a Sublicensee pays more than such fair market value for equity or debt securities then the portion in excess of fair market value shall be considered Non-Royalty Sublicensing Income; and (iii) payments to Company for the purposes of funding the costs of future bona fide documented research of a Licensed Product to be conducted by the Company. The term "fair market value" means the cash consideration which Company, its Affiliates or Sublicensees would realize from an unaffiliated, unrelated buyer in an arm's length sale of an identical item sold in the same quantity and at the same time and place of the transaction. Fair market value will be mutually determined in good faith by Wistar and Company.

Exclusive License Agreement Wistar/OncoCyte

- 1.27 "Past Patenting Costs" has the meaning set forth in Section 7.1.1.
- 1.28 "<u>Patent Challenge</u>" means any direct or indirect dispute, challenge, or assistance in the challenge of the validity, patentability, scope, construction, enforceability, non-infringe-ment or Wistar's ownership of any issued patent comprising the Licensed Patents or any claims thereof, or opposition or assistance in the opposition of the grant of any letters patent comprising the Licensed Patents, in any legal or, administrative proceedings, including in a court of law, before the United States Patent and Trademark Office or other agency or tribunal in any jurisdiction, or in arbitration.
 - 1.29 "Patent Term Extension" has the meaning set forth in Section 7.5.
 - 1.30 "Performance Milestones" has the meaning set forth in Section 4.2.
 - 1.31 "Performance Milestone Dates" has the meaning set forth in Section 4.2.
- 1.32 "Patenting Costs" means any past or ongoing costs incurred or to be incurred, including government fees and attorneys' fees, in the course of Prosecuting the Licensed Patents.
- 1.33 "Proposed Product" means an actual or potential Licensed Product that is for an application, product, sub-field or indication in the Licensed Field, but for which Wistar reasonably believes a Licensed Product is not being actively developed or commercialized by Company, its Affiliates or Sublicensees.
 - 1.34 "Proposed Product Election Notice" has the meaning set forth in Section 2.5.2.
 - 1.35 "Proposed Product Notice" has the meaning set forth in Section 2.5.1.
 - 1.36 "Proposed Product Sublicense" has the meaning set forth in Section 2.5.4.
 - 1.37 "Proposing Third Party" has the meaning set forth in Section 2.5.1.
- 1.38 "Prosecution," "Prosecute," "Prosecuted" or "Prosecuting" shall mean preparation, filing, prosecution, issuance and maintenance of the Licensed Patents, including continuations, divisionals, extensions, re-examinations, reissues, supplemental examination, appeals, interferences, derivation proceedings, oppositions, all other proceedings before the United States Patent and Trademark Office (including the Patent Trial and Appeal Board) and foreign patent offices, and any judicial or other appeals of the foregoing.

Exclusive License Agreement Wistar/OncoCyte

- 1.39 "Person" or "Persons" means any corporation, partnership, joint venture or any other entity or any natural person.
- 1.40 "<u>Regulatory Approval</u>" shall mean, with respect to any jurisdiction, any and all approvals (including appropriate pricing and reimbursement approvals), product and/or establishment licenses, registrations or authorizations of any appropriate regulatory agency, department, bureau or other governmental entity, necessary for marketing a Licensed Product in such jurisdiction, as applicable.
 - 1.41 "Royalty Term" has the meaning set forth in Section 3.3.
- 1.42 "Sublicense" means an agreement in which Company (i) grants or otherwise transfers any of the rights licensed to Company hereunder or other rights that are relevant to designing, developing, testing, making, using, or selling of Licensed Products, (ii) agrees not to assert such rights or to sue, prevent or seek a legal remedy for the practice of same, or (iii) is under an obligation to grant, assign or transfer any such rights or non-assertion, or to forebear from granting or transferring such rights to any other entity, including by means of an option. Agreements expressly considered Sublicenses include licenses, option agreements, "lock up" agreements, right of first refusal agreements, or similar agreements. For the avoidance of doubt, any Proposed Product Sublicense shall be a Sublicense as such term is used hereunder.
 - 1.43 "Sublicensee(s)" means any non-Affiliate third party to which Company has granted a Sublicense.
- 1.44 "<u>Technical Information</u>" means technical information, know-how, data, copyrights, techniques, and other information directly related to the Licensed Products that is not otherwise obligated to another third-party and (i) made, identified or compiled by Dr. Louise C. Showe or laboratory personnel working directly under his/her supervision, (ii) owned and Controlled by Wistar, (iii) in Wistar's possession as of the Effective Date, and (iv) provided to Company by Wistar or Dr. Showe during the Term, including but not limited to the Technical Information listed on Exhibit B attached hereto. Technical Information shall exclude any of the foregoing that are included in a Valid Claim of the Licensed Patents or any other patents or patent applications.
- 1.45 "<u>Term</u>" means the term of this Agreement, which shall commence on the Effective Date and shall remain in effect until the expiration of the Royalty Term in all countries in the Territory with respect to all Licensed Products, unless earlier terminated in accordance with the provisions of this Agreement.
 - 1.46 "<u>Territory</u>" means worldwide.
 - 1.47 "Third Party Proposed Product" has the meaning set forth in Section 2.5.1.

Exclusive License Agreement Wistar/OncoCyte

- 1.48 "Valid Claim" means a claim of (i) a patent application included in the Licensed Patents that has been neither abandoned nor pending for more than [**] ([**]) years or (ii) an issued Licensed Patent that has not been withdrawn, canceled or disclaimed or held invalid by a court or governmental authority of competent jurisdiction in an unappealed or unappealable decision no longer subject to discretionary review (for example, by way of writ of certiorari) or other review.
 - 1.49 "Wistar" has the meaning set forth in the Preamble.
 - 1.50 "Wistar Confidential Information" has the meaning set forth in Section 8.1.1.
 - 1.51 "Wistar Proposed Product" has the meaning set forth in Section 2.5.1.
 - 1.52 "<u>Withholding Taxes</u>" has the meaning set forth in Section 3.10.2.

ARTICLE 2 - GRANT OF LICENSE

2.1 <u>Grant of License.</u> Subject to the terms and conditions contained in this Agreement, Wistar hereby grants to Company an exclusive, royalty-bearing, non-transferable (except in accordance with Section 10.1), sublicensable (solely to the extent permitted by Section 2.4) license under the Licensed Patents and Technical Information solely to make, have made, use, sell, offer for sale, and import the Licensed Products in the Licensed Field in the Territory during the Term.

2.2 Retained Rights.

2.2.1 <u>Reservation by Wistar</u>. Notwithstanding anything to the contrary in this

Agreement, Wistar reserves the right to (i) make, use, practice and further develop the Licensed Patents and Technical Information for educational, research, and other internal purposes; (ii) grant to any academic, government, research or non-profit institution or organization the right to make, use and practice the Licensed Patents or Technical Information for non-commercial research and educational purposes; and (iii) grant licenses under the Licensed Patents or Technical Information to any party for any field, product, service or territory other than the Licensed Products in the Licensed Field in the Territory for so long as Company has an exclusive license to Licensed Products in the Licensed Field in the Territory. Company shall have no right or license in or to any research conducted pursuant to (i), (ii) or (iii) above or any results, inventions or discoveries generated thereby.

Funding Agencies Interest. Notwithstanding anything to the contrary in this Agreement, any and all licenses and other rights granted hereunder are limited by and subject to the rights and requirements of (a) the United States Government under any law or agreement, including rights and requirements which may attach as a result of United States Government sponsorship of research in connection with which an invention covered by the Licensed Patents was conceived or first actually reduced to practice, as set forth in 35 U.S.C. §\$200-212, and 37 C.F.R. Part 401, or any successor statutes or regulations, and in relevant United States Government research grants or contracts with Wistar, as such rights and requirements may be amended or modified by law, rule or regulation and which include (i) the grant of a nonexclusive, nontransferable, irrevocable, paid-up license to practice or have practiced for or on behalf of the United States Government any of the Licensed Patents throughout the world (as set forth in 35 U.S.C. §202(c)(4)) and (ii) the requirement that Licensed Products used or sold in the United States shall be manufactured substantially in the United States (as set forth in 35 U.S.C. §204); (b) any local, state or philanthropic funding agencies or entities under any contract, grant or similar agreement with such agency or entity (collectively the "Funding Agencies Interest").

Exclusive License Agreement Wistar/OncoCyte

- 2.3 <u>No Rights or Licenses by Implication</u>. No rights or licenses with respect to the Licensed Patents, Technical Information, or otherwise are granted or deemed granted hereunder or in connection herewith, other than those rights or licenses expressly granted in this Agreement.
- 2.4 <u>Right to Sublicense</u>. Company shall have the right to sublicense to any third party the rights conferred upon Company under this Agreement, subject to the following conditions:
- 2.4.1 Wistar shall have the right to approve in advance any Sublicensee if Company is not selling Licensed Product at the time of sublicensing negotiations, provided that such approval shall not be unreasonably withheld; and further provided, however, that such approval shall be deemed to have been given if Wistar does not object to the proposed Sublicensee within [**] ([**]) business days after Company notifies Wistar in writing of the name of such Sublicensee.
- 2.4.2 Any Sublicense shall be in writing, shall be consistent with all of the terms and conditions of this Agreement, and shall incorporate terms and conditions sufficient to enable Company to comply with this Agreement. Without limiting the foregoing, each Sublicense shall (i) provide that in the event Sublicensee brings a Patent Challenge against Wistar or assists another party in bringing a Patent Challenge against Wistar (except as required under a court order or subpoena or if legally compelled by an administrative agency) then Company may terminate the Sublicense, (ii) require Sublicensee to indemnify, hold harmless and defend Wistar and carry insurance under the same terms set forth in Article 6 below, and (iii)state that Wistar is an intended third party beneficiary of such Sublicense, including for the purpose of enforcing such termination, indemnification, and insurance provisions.
- 2.4.3 No Sublicensee shall be permitted to sublicense further any of its rights under any Sublicense. Each Sublicense shall contain an agreement and acknowledgment by the Sublicensee that such Sublicensee and the Sublicensee are subject to the terms and conditions of the license granted to Company under this Agreement.
- 2.4.4 Notwithstanding any Sublicense, Company shall remain primarily liable to Wistar for all of Company's duties and obligations contained in this Agreement, and any act or omission of a Sublicensee which would be a breach of this Agreement if performed by Company shall be deemed to be a breach by Company of this Agreement.

Wistar Reference No. LIC15-35

Page 8 of 32

Exclusive License Agreement Wistar/OncoCyte

Certain information has been omitted under a request for confidential treatment, and the omitted information has been filed with the Commission
Confidential portions are marked [**].

- 2.4.5 If Wistar has a claim arising under this Agreement against a Sublicensee, Wistar may seek a remedy directly against Company and may, but is not required to, seek a remedy against the Sublicensee.
- 2.4.6 If Company becomes subject to a Bankruptcy Event, all payments then or thereafter due and owing to Company from its Sublicensees shall thereupon, and without any notice from Wistar to any such Sublicensee, become payable directly to Wistar for the account of Company; provided, however, that Wistar shall remit to Company any amount by which such payments exceed the amounts owed by Company to Wistar.
- 2.4.7 Company shall furnish Wistar with a fully executed copy of any Sublicense agreement within thirty (30) days after execution without redaction.
 - 2.4.8 Any sublicense that is not in compliance with all of the provisions of this Section 2.4 shall be void.

2.5 <u>Wistar or Third Party Proposed Products.</u>

- 2.5.1 Notice of Proposed Products. If at any time after the Effective Date, (i) Wistar identifies a Proposed Product, or (ii) a third party ("Proposing Third Party") identifies a Proposed Product and makes a bona fide proposal to Wistar for the development of such Proposed Product, and, in each case (i) or (ii), Wistar is interested in having such Proposed Product developed, then Wistar may give written notice thereof to Company ("Proposed Product Notice"), which notice shall include information regarding the relevant application, product, sub-field or indication. Wistar shall not be required to include in any Proposed Product Notice (a) any information that is subject to restrictions of confidentiality with any third party, or (b) any information that originates with Wistar personnel who do not assent to its disclosure to Company. For the purposes of this Agreement, (A) a Proposed Product identified by a third party as described in (i) above shall be a "Wistar Proposed Product," and (B) a Proposed Product identified by Wistar as described in (ii) above shall be a "Third Party Proposed Product."
- 2.5.2 <u>Election by Company.</u> Within [**] ([**]) days following Company's receipt of the Proposed Product Notice, Company shall provide Wistar written notice stating whether or not Company elects to develop and commercialize such Proposed Product on its own ("<u>Proposed Product Election Notice</u>"). If Company does not timely provide a Proposed Product Election Notice, Wistar shall be entitled to exclude such Proposed Product from the license hereunder to Company, or convert such license to a non-exclusive license with respect to such Proposed Product, and Wistar shall be free to further develop such Proposed Product internally or license the Licensed Patents, on an exclusive or non-exclusive basis, as the case may be, to any third party for the development and commercialization of such Proposed Product.

2.5.3 <u>Internal Development</u>.

(i) If Company timely provides a Proposed Product Election Notice and affirmatively elects to develop and commercialize the Proposed Products on its own, then (a) the Proposed Product Election Notice shall include a development plan and milestones for the development and commercialization of such Proposed Product that have been approved by the Board of Directors of Company and (b) the parties shall negotiate in good faith to agree on the development plan and milestones applicable to such Proposed Product.

Exclusive License Agreement Wistar/OncoCyte

- (ii) If the parties agree on a development plan and milestones, Company shall maintain its exclusive license with respect to such Proposed Product, but shall be obligated to (a) use commercially reasonable efforts to develop and commercialize the Proposed Product in accordance with such new development plan and (b) meet the milestones with respect to the Proposed Product. If (A) the parties do not agree on a development plan and milestones that are acceptable to Wistar, in its reasonable judgment, or (B) the parties agree on such a development plan and milestones, but Company thereafter fails to comply in any material respect with such mutually agreed development and commercialization obligations, Wistar shall be entitled from such point onward to exclude such Proposed Product from the license hereunder to Company, or convert such license to a non-exclusive license with respect to such Proposed Product, and Wistar shall be free to grant to third parties licenses under the relevant Licensed Patents or Technical Information, on an exclusive or non-exclusive basis, as the case may be, to develop and commercialize such Proposed Product.
- 2.5.4 <u>0</u>. If Company declines to develop and commercialize such Proposed Product on its own, then the Proposed Product Election Notice shall indicate whether or not Company wishes to provide a Sublicense for the development and commercialization of such Proposed Product ("Proposed Product Sublicense"), which Sublicense shall be with (i) the Proposing Third Party, in the case of a Third Party Proposed Product, or (ii) a third party reasonably acceptable to Wistar, in the case of a Wistar Proposed Product. If Company elects to provide a Proposed Product Sublicense, it shall do so in good faith within [**] ([**]) days of provision of the Proposed Product Election Notice. Such Proposed Product Sublicense will contain reasonable financial terms taking into account the field/indication and Company's obligations to Wistar, and reasonable non-financial terms consistent with Company's obligations to Wistar. If (a) Company elects not to provide such Proposed Product Sublicense, (b) such Proposed Product Sublicense has not been entered within [**] ([**]) days of the Proposed Product Election Notice, or (c) Company enters such Proposed Product Sublicense, but Company thereafter fails to comply in any material respect with such Proposed Product Sublicense, Wistar shall be entitled to exclude such Proposed Product from the license hereunder to Company, or convert such license to a non-exclusive license with respect to such Proposed Product, and Wistar shall be free to (A) in the case of a Third Party Proposed Product, license the Licensed Patents to the Proposed Product, further develop the Wistar Proposed Product internally, and/or license the Licensed Patents or Technical Information, on an exclusive or non-exclusive basis, as relevant to any third party for the development and commercialization of such Proposed Product.
- 2.5.5 <u>Improvements</u>. Wistar shall use good faith efforts to inform Company of new improvements directly related to the Licensed Patents and Technical Information invented by Dr. Louise C. Showe during the term of this Agreement, unless such disclosure by Wistar is otherwise precluded or prohibited by the terms of an agreement with a third party.

Exclusive License Agreement Wistar/OncoCyte

2.5.6 <u>Sponsored Research Agreement.</u> The terms of Section 5.3 of that certain amended Sponsored Research Agreement dated as of September 18, 2013 by and between Wistar and Company (the "<u>SRA</u>"), shall remain in effect as long as the SRA remains in effect.

ARTICLE 3 - PAYMENTS

- 3.1 <u>License Fee</u>. On the Effective Date, Company shall pay to Wistar a nonrefundable, non-creditable license issue fee of fifty thousand dollars (\$50,000).
- 3.2 <u>Royalties</u>. Company shall pay to Wistar a running royalty on Net Sales of all Licensed Products in accordance with the table set forth below within thirty (30) days following the last day of the Calendar Quarter in which such royalty accrues:
 - 3.2.1 For Licensed Products as defined in Section 1.24:

Royalty Percentage	for that portion of cumulative Net Sales of Licensed Products:
5%	Up to \$[**]
4%	Greater than \$[**] and less than \$[**]
3%	\$[**] and greater

- 3.2.2 If Company is notified that the manufacture or sale of Licensed Products, in a particular country, infringes intellectual property rights of any third party in such country, the Company will inform Wistar of said notification in writing within [**] ([**]) calendar days of receipt. In the event Company is paying Wistar royalties under Section 3.2.2, and is required to enter into a bona fide license agreement pursuant to which Company pays a third party royalties directly related to Licensed Products in order to continue the manufacture or sale of Licensed Products, Wistar agrees to reduce the earned royalties on Net Sales of Licensed Products due to Wistar under Section 3.2.2 in the specified country, by the amount of such third party royalties actually paid by Company, provided, however, that Wistar's earned royalties on Net Sales in the specified country shall not be reduced by more than fifty percent (50%) during any such payment period.
- 3.3 <u>Royalty Term.</u> The period during which the royalties set forth in Section 3.2 shall be payable, on a Licensed Product-by-Licensed Product and country-by-country basis, shall commence on the Effective Date and continue until the later of (i) the date a Valid Claim of a Licensed Patent covering the making, having made, use, sale, offering for sale or importation of such Licensed Product no longer exists, or (ii) the tenth (10th) anniversary of the First Commercial Sale of such Licensed Product in each country (the "Royalty Term").

3.4 <u>Minimum Annual Royalties</u>.

3.4.1 Company shall pay to Wistar as a non-refundable but creditable advance against royalties during the ensuing year, minimum annual royalties on the following dates in the corresponding amounts (the "Minimum Annual Royalties"):

Exclusive License Agreement Wistar/OncoCyte

- (i) [**] dollars (\$[**]) due on January 1 following the First Commercial Sale of a Licensed Product; and
- (ii) [**] dollars (\$[**]) due on January 1 of each year of the Term thereafter.
- 3.4.2 Minimum Annual Royalties shall be available for credit against royalties only during the year in which such Minimum Annual Royalties are paid and shall not be available for credit in any other year.
- 3.5 Non-Royalty Sublicensing Income. Company shall pay Wistar [**] percent ([**]%) of Non-Royalty Sublicensing Income within [**] ([**]) days following the last day of the Calendar Quarter in which such Non-Royalty Sublicensing Income accrues.
- 3.6 <u>Maintenance Fees.</u> Wistar shall waive the annual license maintenance fee so long as Company has initiated sales of Licensed Products no later than January 1, 2018. In the event Company fails to launch first commercial sale of Licensed Product by January 1, 2018, Company shall pay to Wistar an annual license maintenance fee of [**] dollars (\$[**]) to be paid on each anniversary of the Effective Date.
- 3.7 <u>Milestone Payments</u>. Company shall pay to Wistar milestone payments in the following amounts within [**] ([**]) days after the occurrence of each the following events:
 - 3.7.1 [**] dollars (\$[**]) upon Wistar's first transfer of the [**] list to Company;
- 3.7.2 [**] dollars (\$[**]) upon the initial IDE Filing or equivalent regulatory filing in the first country by Company or any Sublicensee for each Licensed Product;
- 3.7.3 [**] dollars (\$[**]) upon the first sale of a Licensed Product that is a laboratory developed test performed within a CLIA certified laboratory by Company or any Sublicensee.
- 3.7.4 These milestone payments shall not be credited against or otherwise reduce royalties or other compensation provided for in this Agreement.
 - 3.8 <u>Change of Control</u>. Intentionally omitted and not referenced in this Agreement
- 3.9 Reports. Company shall deliver to Wistar within [**] ([**]) days after the end of each Calendar Quarter a complete and accurate report for that Calendar Quarter certified by a senior financial officer of Company.

Exclusive License Agreement Wistar/OncoCyte

3.10 Payments.

- 3.10.1 All dollar amounts referred to in this Agreement are expressed in United States dollars. Liability for royalties on Licensed Products manufactured by Company or Sublicensee shall accrue when a Licensed Product is sold, used or otherwise disposed of; provided that, if a partial payment is made, a royalty shall accrue pro rata to such partial payment. Any royalties not paid to Wistar when due hereunder shall accrue interest from the due date until paid at a rate equal to the lesser of twelve percent (12%) per annum or the maximum interest rate allowed by applicable law, whichever is greater. Notwithstanding the foregoing and subject to Section 9.2.1, Wistar shall be entitled to treat any such late payment as a material breach of this Agreement, notwithstanding the payment of interest. All payments hereunder shall be payable in U.S. dollars, at Wistar's option, by a Company check payable to the order of "The Wistar Institute of Anatomy and Biology" and drawn on a U.S. bank, or as to any payment for which Wistar has given Company at least fifteen (15) days' written notice prior to the due date of such payment, by wire transfer to Wistar's account at [**]. Where it is necessary to convert the amount of royalties due from another currency into U.S. dollars, conversion shall be made using one of the following rates as published or issued on the last business day of the Calendar Quarter in which such royalties have accrued:
- (i) the spot rate or the mean of the buy and sell spot rates, if no single rate is published, as published by "The Wall Street Journal;" or
- (ii) if no rate(s) is (are) therein published by the "Wall Street Journal," then at the rate prevailing at the close of business at J. P. Morgan Chase & Co., New York, New York; or
- (iii) at the currency conversion rate published or issued at the close of business by a third party selected by Company, provided that Company has obtained Wistar's prior written consent to use such third party rate for calculation of royalties due to Wistar.
- 3.10.2 All payments under this Agreement shall be made without any deduction or withholding for or on account of any tax, except as expressly permitted in this Agreement. If any income or other taxes, withholdings or other deductions required by applicable law to be withheld or deducted from any of the payments made by or on behalf of Company hereunder ("Withholding Taxes") are imposed on a payment by any applicable law, Company shall pay such Withholding Taxes to the proper taxing authority and, if available, evidence of such payment shall be secured and sent to Wistar within [**] ([**]) month of such payment. In the case of any Withholding Taxes imposed with respect to any payment hereunder, Company shall pay to Wistar an additional amount as is necessary to ensure that the amount actually received by Wistar with respect to such payment, free and clear of the Withholding Taxes (including any such Withholding Taxes imposed on such additional amount), shall equal the amount of the payment that would have been made if no such Withholding Taxes had applied.

Wistar Reference No. LIC15-35

Page 13 of 32

Exclusive License Agreement
Wistar/OncoCyte

3.11 Records. Company shall keep, and shall cause its Sublicensees to keep, complete and accurate books and records of all Licensed Products sold which enable the royalties and other amounts payable hereunder to be verified. Upon reasonable prior notice to Company or its Sublicensee and during normal business hours, an auditor paid for and selected by Wistar may inspect such books and records of Company and its Sublicensees for the [**] ([**]) year period immediately preceding the date of inspection to verify the correctness of the reports given to Wistar under Section 3.9. If Wistar's auditor determines that the Company or Sublicensee has underpaid royalties and other amounts payable by [**] percent ([**]%) or more, Company shall pay the costs and expenses of the audit and the right of inspection shall extend to books and records for periods prior to such [**] ([**]) year period. Nothing contained in this Section 3.11 shall shorten the period established by any applicable statute of limitations.

ARTICLE 4 - CERTAIN OBLIGATIONS OF COMPANY

- 4.1 <u>Diligent Efforts</u>. Company, acting itself and/or through its Sublicensees, (i) shall use commercially reasonable, diligent efforts to develop Licensed Products and to bring Licensed Products to market through a thorough, vigorous and diligent program for exploitation of the Licensed Patents and to continue active, diligent marketing efforts for Licensed Products throughout the Term, consistent with sound and reasonable business practices, and (ii) shall endeavor to keep Licensed Products, to the extent commercially reasonable, available to the public.
- 4.2 <u>Performance Milestones</u>. In addition to Company's diligence obligations in Section 4.1, Company, acting itself and/or through its Sublicensees, shall perform or fulfill the following obligations (the "<u>Performance Milestone(s)</u>") by the dates set forth below (the "<u>Performance Milestone Date(s)</u>"):
- 4.2.1 Within thirty (30) days after the Effective Date, Company shall furnish Wistar with a written research and development plan describing the major tasks to be achieved in order to develop and bring to market Licensed Products, a timeline for achievement of such tasks, and an estimate of the number of staff and financial and other resources to be devoted to such development and commercialization effort ("Commercialization Plan"); provided that if Wistar has not objected to such Commercialization Plan within thirty (30) days after receiving it from Company, such Commercialization Plan as so presented by Company, the parties will work together expeditiously and in good faith to attempt to arrive at a mutually agreeable Commercialization Plan.
 - 4.2.2 Company or its permitted Sublicensees shall no later than December 31, 2017 commence commercial sales of a Licensed Product.
- 4.3 <u>Diligence Failure</u>. In the event Wistar determines that Company has failed to fulfill any of its obligations under Sections 4.1 or 4.2, Wistar may terminate this Agreement in accordance with Section 9.2; provided, however, that if Company is meeting its obligations under Section 4.1, Company may purchase up to a total of three (3) one-year extensions of any Performance Milestone under Section 4.2 by paying Wistar, on or before the Performance Milestone Date for such Performance Milestone, twenty-five thousand dollars (\$25,000) the first extension and fifty thousand dollars (\$50,000) for each extension thereafter.

Exclusive License Agreement Wistar/OncoCyte

4.4 <u>Diligence Reports</u>.

- 4.4.1 Company shall provide Wistar on December 1 of each year with written reports, setting forth in such detail as Wistar may reasonably request, the progress of the development, evaluation, testing and commercialization of the Licensed Products, including information on (i) the progress of matters related to Regulatory Approvals and (ii) progress made toward the objectives set forth in the Commercialization Plan, including the progress of any Sublicensees developing Licensed Products. Company also shall notify Wistar within thirty (30) days after the First Commercial Sale of a Licensed Product by Company or any Sublicensee.
- 4.4.2 In order that Wistar may provide the United States Department of Health and Human Services with information required under the Federal Government Interest, Company shall, at its cost, prepare and provide written annual reports to Wistar containing sufficient information to enable the Department of Health and Human Services to evaluate Company's progress in the development of Licensed Products, but shall not contain information considered proprietary or confidential by Company. Such reports shall be provided no later than January 1 each year.

4.5 <u>Compliance with Laws</u>.

- 4.5.1 Company shall comply with all applicable laws, rules and regulations pertaining to the development, testing, manufacture, marketing and import or export of the Licensed Products. Without limiting the foregoing, Company acknowledges that the transfer and use by foreign nationals of certain commodities and technical data is subject to United States laws and regulations controlling the export and use by foreign nationals of such commodities and technical data, including the Arms Export Control Act, the International Traffic in Arms Regulations ("ITAR"), the Export Administration Regulations ("EAR") and the laws and regulations implemented by the Office of Foreign Assets Control, U.S. Department of the Treasury ("OFAC"). These laws and regulations, among other things, prohibit or require a license for the export or use by foreign nationals of certain types of technical data to specified countries. Company shall comply with all United States laws and regulations controlling the export or use by foreign nationals of such commodities and technical data, including ITAR, EAR, and OFAC, and shall be solely responsible for any violation thereof by Company or its Sublicensees.
- 4.5.2 To the extent required by the Funding Agencies Interest, all Licensed Products to be used or sold in the United States shall be manufactured substantially in the United States, and Company shall take such actions as are necessary to assure that it and its Sublicensees comply with the obligations imposed by this Section 4.5.2.

Exclusive License Agreement Wistar/OncoCyte

- 4.6 <u>Conflict of Interest</u>. Company acknowledges that Wistar's employees and staff members and the employees and staff members of Wistar's Affiliates are subject to the applicable policies of Wistar and such Affiliates, including policies regarding conflicts of interest, intellectual property and other matters. Company shall not enter into any oral or written agreement with such employee or staff member which conflicts with any such policy.
- 4.7 <u>Patent Notices</u>. Company shall mark the Licensed Products sold in the United States with all applicable patent numbers. All Licensed Products shipped to and/or sold in other countries shall be marked and labeled in such a manner as to conform with all applicable laws of the country where the Licensed Products are sold.
 - 4.8 <u>Regulatory Approvals</u>. Company shall be responsible for obtaining and maintaining, at its cost and expense, all Regulatory Approvals.

ARTICLE 5 - REPRESENTATIONS AND WARRANTIES

- 5.1 <u>Representations and Warranties.</u>
- 5.1.1 Company represents and warrants that it is a corporation, duly organized, validly existing and in good standing under the laws of California, and has all requisite corporate power and authority to execute, deliver and perform this Agreement; and Wistar represents and warrants that it is a nonprofit corporation, duly organized, validly existing and in good standing under the laws of the Commonwealth of Pennsylvania, and has all requisite corporate power and authority to execute, deliver and perform this Agreement; and
- 5.1.2 Company and Wistar each represent and warrant to the other that this Agreement, when executed and delivered by a party to this Agreement, shall be the legal, valid and binding obligation of such party, enforceable against such party in accordance with its terms.

ARTICLE 6 - LIMITATION ON LIABILITY AND INDEMNIFICATION

6.1 <u>Limitation on Liability.</u> IN NO EVENT SHALL WISTAR, ITS TRUSTEES, MANAGERS, OFFICERS, AGENTS, EMPLOYEES, FACULTY, AFFILIATED INVESTIGATORS, PERSONNEL OR STAFF BE LIABLE TO COMPANY, ITS SUCCESSORS, ASSIGNS, SUBLICENSEES OR ANY OTHER PERSON OR ENTITY FOR ANY LOSS OF PROFITS, LOSS OF BUSINESS, INTERRUPTION OF BUSINESS, OR FOR ANY INDIRECT, SPECIAL OR CONSEQUENTIAL DAMAGES OF ANY KIND, WHETHER UNDER THIS AGREEMENT OR OTHERWISE, EVEN IF SUCH PERSON HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH LOSS.

Exclusive License Agreement
Wistar/OncoCyte

- 6.2 No Warranties. THE LICENSED PATENTS AND TECHNICAL INFORMATION ARE PROVIDED ON AN "AS IS" BASIS AND WISTAR MAKES NO REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED, WITH RESPECT TO THE LICENSED PATENTS, TECHNICAL INFORMATION OR ANY MATERIALS DERIVED THEREFROM OR ANY LICENSED PRODUCTS, INCLUDING REPRESENTATIONS OR WARRANTIES OF COMMERCIAL UTILITY, MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE, VALIDITY OR ENFORCEABILITY OF THE LICENSED PATENTS, OR THAT THE USE OF THE LICENSED PATENTS, TECHNICAL INFORMATION OR ANY MATERIALS DERIVED THEREFROM OR ANY LICENSED PRODUCTS WILL NOT INFRINGE ANY PATENT, COPYRIGHT OR TRADEMARK OR OTHER PROPRIETARY OR PROPERTY RIGHTS OF OTHERS. WISTAR EXPRESSLY DISCLAIMS ANY WARRANTY THAT THE LICENSED PATENTS OR TECHNICAL INFORMATION ARE FREE FROM THE RIGHTFUL CLAIMS OF ANY THIRD PARTY. WISTAR SHALL NOT BE LIABLE TO COMPANY, ITS SUCCESSORS, ASSIGNS, SUBLICENSEES OR ANY THIRD PARTY WITH RESPECT TO ANY CLAIM ON ACCOUNT OF, OR ARISING FROM, THE USE OF THE LICENSED PATENTS OR TECHNICAL INFORMATION PROVIDED HEREUNDER OR THE MANUFACTURE, USE OR SALE OF LICENSED PRODUCTS OR ANY OTHER MATERIAL OR ITEM DERIVED THEREFROM.
- 6.2.1 For clarity, to the knowledge of the current staff of the Office of Business Development of Wistar, no third parties have asserted that the Licensed Products infringe any third party patent rights.
- Indemnification. Company shall indemnify and hold harmless Wistar, its trustees, managers, officers, agents, employees, faculty, affiliated investigators, personnel and staff (collectively and individually, the "Indemnified Parties" or "Indemnified Party"), from and against any and all liability, loss, damage, action, claim or expense (including attorney's fees) suffered or incurred by the Indemnified Parties due to claims by a Person not a party to this Agreement (individually, a "Liability" and collectively, the "Liabilities") which result from or arise out of (a) the license granted hereunder and any Sublicense granted pursuant thereto, (b) the development, use, manufacture, promotion, sale or other disposition of the Licensed Patents, Technical Information or any Licensed Products by Company, its assignees, Sublicensees, Third Party Transferees, vendors or other third parties, (c) the breach of any representation, warranty, or covenant of this Agreement by Company, or of a Sublicense by any Sublicensee, or (d) the successful enforcement by an Indemnified Party of its rights under this Section 6.3. Subject to Section 6.3.4, this indemnification obligation shall apply regardless of the negligence of the Indemnified Party. Without limiting the foregoing, Company shall indemnify and hold harmless the Indemnified Parties from and against any Liabilities resulting from:
- 6.3.1 any product liability or other claim of any kind related to the use of a Licensed Product manufactured, sold or otherwise disposed of by Company, its assignees, Sublicensees, vendors or other third parties;
- 6.3.2 any claim that the Licensed Patents or the design, composition, manufacture, use, sale or other disposition of any Licensed Product infringes or violates any patent, copyright, trademark or other intellectual property rights of any third party; or
- 6.3.3 Clinical Trials or studies conducted by or on behalf of Company or any Sublicensee relating to the Licensed Products, including any claim by or on behalf of a human subject of any such Clinical Trial or study, any claim arising from the procedures specified in any protocol used in any such Clinical Trial or study, any claim of deviation, authorized or unauthorized, from the protocols of any such Clinical Trial or study, and any claim resulting from or arising out of the manufacture or quality control by a third party of any substance administered in any Clinical Trial or study.

Exclusive License Agreement Wistar/OncoCyte

- 6.3.4 Notwithstanding the foregoing, Company shall not be obligated to indemnify and hold harmless the Indemnified Parties from and against any Liabilities that result from or arise out of an Indemnified Party's gross negligence or willful misconduct.
- 6.4 Procedures. The Indemnified Party shall promptly notify Company of any claim or action giving rise to a Liability subject to the provisions of Section 6.3, and Wistar and Company then shall confer promptly concerning the claim or action and the response thereto. Unless the parties otherwise agree in writing, Company shall defend any such claim or action, at its cost and expense, with counsel reasonably satisfactory to Wistar. Company shall not settle or compromise any such claim or action in a manner that (i) imposes any restrictions or obligations on any Indemnified Party without such Indemnified Party's written consent, not to be unreasonably withheld, or (ii) grants any rights to the Licensed Patents or Technical Information without Wistar's written consent, not to be unreasonably withheld. If Company fails or declines to assume the defense of any such claim or action within [**] ([**]) days after notice thereof, Wistar may assume the defense of such claim or action for the account and at the risk of Company, and any Liability related thereto shall be conclusively deemed a Liability of Company. Company shall pay promptly to the Indemnified Party any Liabilities to which the foregoing indemnity related, as incurred. The indemnification rights of Wistar and any other Indemnified Party contained herein are in addition to all other rights which Wistar or such Indemnified Party may have at law or in equity or otherwise.
 - 6.5 <u>Insurance</u>. Company shall maintain general liability and product liability insurance as follows:
- 6.5.1 Beginning with the Effective Date and for [**] ([**]) years after the date of expiration or termination of this Agreement, general liability insurance in amounts not less than [**] dollars (\$[**]) per incident and [**] dollars (\$[**]) in the aggregate.
- 6.5.2 Beginning with the commencement of human Clinical Trials and for [**] ([**]) years after the date of expiration or termination of this Agreement, product liability insurance in amounts not less than [**] dollars (\$[**]) per incident and [**] dollars (\$[**]) in the aggregate.
- 6.5.3 Such insurance shall be issued by an insurance company rated [**] or better and naming Wistar as an additional insured. The minimum insurance amounts specified herein shall not be deemed a limitation on Company's indemnification liability under this Agreement. Company shall provide Wistar with copies of endorsements to such policies. Company shall notify Wistar at least thirty (30) days prior to cancellation of any such coverage. To the extent Company is awarded a business interruption insurance award which provides for lost profits, Company shall pay to Wistar reasonable royalties for the period of the award which payment shall be based upon projections of Net Sales of Licensed Products and the history of royalties paid hereunder for such Net Sales.

Exclusive License Agreement Wistar/OncoCyte

6.5.4 Company shall require any Sublicensee to maintain insurance under the same terms as set forth in Sections 6.5.1 and 6.5.2 above, including naming Wistar as an additional insured.

ARTICLE 7 - PATENTS AND INFRINGEMENT

7.1 Prosecution of Patents.

- 7.1.1 Wistar shall have the exclusive responsibility and control over the Prosecution of the Licensed Patents. Company shall reimburse Wistar for Patenting Costs incurred by Wistar prior to the Effective Date ("Past Patenting Costs") within [**] ([**]) days of the Effective Date. With respect to any Patenting Costs incurred by or on behalf of Wistar after the Effective Date, Company shall remit payment of such Patenting Costs within [**] ([**]) days after Company receives invoices for same. Notwithstanding the foregoing, at least [**] ([**])) days before a particular action is required for the protection of certain rights comprising the Licensed Patents (the "Bar Date"), Wistar shall have the right to request advance payment of reasonable estimated Patenting Costs for such action if such estimated Patenting Costs are at least [**] dollars (\$[**]), and Company shall be obligated to pay the amount of such estimated Patenting Costs no less than [**] ([**]) days before the Bar Date. So long as Wistar's request is timely made, Wistar shall have no obligation to take or have taken such action, and no liability for failing to take such action, to protect the Licensed Patents at issue, unless the estimated Patenting Costs are timely paid by Company, even if the result is the irrevocable loss of rights.
- 7.1.2 Company and Wistar shall mutually determine the jurisdictions, other than the United States, where the Licensed Patents shall be Prosecuted. If Company declines to pay for Patenting Costs in any jurisdiction, Wistar may do so at its cost and expense but such patents and the subject matter of any application relating thereto shall be excluded from the definition of Licensed Patents.
- 7.1.3 If Wistar elects not to Prosecute any patent or patent application included in the Licensed Patents, it shall notify Company at least [**] ([**]) days prior to taking, or not taking, any action which would result in abandonment, withdrawal, or lapse of such patent or patent application. Company shall then have the right to Prosecute such patent or patent application at its own cost and expense in Wistar's name.
- 7.1.4 Each party shall cooperate with the other party to execute all lawful papers and instruments and to make all rightful oaths and declarations as may be necessary in the Prosecution of all such patents and other applications and protections referred to in this Article 7.
- 7.1.5 All non-public information exchanged between the parties or between Wistar's patent counsel and Company regarding Prosecution and enforcement of the Licensed Patents, and all shared information regarding analyses or opinions of third party intellectual property, shall be deemed Confidential Information. In addition, the parties acknowledge and agree that, with regard to Prosecution and enforcement of the Licensed Patents, the interests of the parties as licensor and licensee are to obtain the strongest patent prosecution possible, and as such, are aligned and are legal in nature. The parties agree and acknowledge that they have not waived, and nothing in this Agreement constitutes a waiver of, any legal privilege concerning the Licensed Patents or the Confidential Information, including privilege under the common interest doctrine and similar or related doctrines.

Exclusive License Agreement Wistar/OncoCyte

- 7.2 <u>Ownership</u>. Wistar shall retain all right, title and interest in and to the Licensed Patents and Technical Information regardless of which party Prosecutes the patents, subject to the express license granted to Company under Article 2 hereof.
- 7.3 <u>Infringement</u>. Each party shall promptly notify the other party in writing of any infringement or possible infringement of any Licensed Patent in the Licensed Field. Company shall have the first right, but not the obligation, to prosecute such infringement in the Licensed Field at its own expense. Company shall use the same degree of diligence in prosecuting such infringement as it uses or would use in prosecuting infringement of its own patent rights.
- 7.3.1 In the event Company elects to prosecute such infringement, Wistar shall cooperate with Company, at Company's reasonable request and Company's sole expense, in any such infringement action. Company shall reimburse Wistar for any such expense within [**] ([**]) days after Company's receipt of invoice for the same. Company shall keep Wistar informed of the status and progress of any action brought under this Section 7.3.1 and Company shall not settle or compromise any such suit in a manner that [**], without Wistar's written consent. Prior to commencing any such infringement action, Company shall consult with Wistar and shall consider the views of Wistar regarding the advisability of the proposed action and its effect on the public interest. [**].
- 7.3.2 If Company fails to prosecute such infringement within [**] ([**]) days after receiving notice thereof, Wistar shall have the right, but not the obligation, to prosecute such infringement at its own expense. In such event, Company shall cooperate with Wistar, at Wistar's expense. Wistar shall not settle or compromise any such suit in a manner that imposes any limitations or restrictions on the rights granted to Company in Article 2 hereof without Company's written consent. [**].
- 7.3.3 Any recovery obtained by the prosecuting party as a result of such proceeding, by settlement or otherwise, shall be applied first to the prosecuting party, an amount equal to its costs and expenses of the litigation, with the remainder to be paid [**].
- 7.4 <u>Certain Notices</u>. Company shall notify Wistar at least [**] ([**]) days before the Company uses or exports the Licensed Patents or any Licensed Product in or to any country outside the United States to allow Wistar to make any patent filings or to take other actions necessary to protect the Licensed Patents.

Exclusive License Agreement Wistar/OncoCyte

- 7.5 Patent Term Extension Obligations. Company shall keep Wistar fully informed of Company's and each Sublicensee's progress toward Regulatory Approval for commercial sale of each Licensed Product with respect to each Licensed Patent hereunder. Company shall assist Wistar in determining with respect to such Licensed Products if the Licensed Patents would be eligible for patent term extension pursuant to 35 U.S.C. §§156, and, as appropriate, applicable foreign patent laws (a "Patent Term Extension"). [**]. [**]. At Wistar's request, Company shall, in a timely manner, assist Wistar in preparing an application for Patent Term Extension in compliance with 35 U.S.C. §156 et seq., and, as appropriate, any applicable foreign patent laws. Company and its Sublicensees shall cooperate fully with Wistar in preparing the applications for Patent Term Extension. Company agrees to join in such applications at Wistar's request. Company shall fully support such applications and shall provide such information as may be requested in support of such applications by Wistar or by the government.
- Licensed Patent Challenges. In the event that Company or a Sublicensee or any of their Affiliates directly or indirectly brings, or assists in 76 bringing, a Patent Challenge, then (i) Company shall provide Wistar with at least sixty (60) days' notice prior to taking any such action, (ii) the parties consent that Section 10.9 shall apply; (iii) Company shall pay all costs, fees and expenses associated with such Patent Challenge that are incurred by Wistar and its trustees, managers, officers, agents, employees, faculty, affiliated investigators, personnel, and staff, including attorneys' fees and all costs associated with administrative, judicial or other proceedings, within thirty (30) days after receiving an invoice from Wistar for same; (iv) the exclusive licenses granted in this Agreement shall, as of the date of initiation of said challenge or opposition, automatically convert to a non-exclusive license for the remainder of the Term, and Wistar shall have the right to grant licenses under the Licensed Patents to third parties, subject to the then-existing non-exclusive license provided herein; (v) any fees, royalties, milestones or revenues payable to Wistar under Sections 3.1 through 3.7 shall double in amount if and when any Licensed Patent survives the Patent Challenge such that it remains valid in whole or in part; and (vi) at any time after the Patent Challenge is brought, Wistar may, at its option, terminate this Agreement according to Section 9.2; provided that if any of subsections (i) through (vi) are held invalid or unenforceable for any reason, such invalidity or unenforceability shall not affect any of the other said subsections. Notwithstanding any provision of this Agreement to the contrary, Company shall not have the right to assume or participate in the defense, settlement or other disposition of such Patent Challenge, but shall pay associated costs, fees and expenses as provided in subsection (iii) above. The parties agree any challenge or opposition to a Licensed Patent by Company may be detrimental to Wistar, and that the above provisions shall constitute reasonable liquidated damages to reasonably compensate Wistar for any loss it may incur as a result of Company taking such action.

ARTICLE 8 - CONFIDENTIALITY

8.1 <u>Confidentiality</u>

8.1.1 "<u>Wistar Confidential Information</u>" means (i) the Technical Information; (ii) any information provided to Company in connection with Prosecution under this Agreement, (iii) any information or material in tangible form that is marked as "confidential" or proprietary by Wistar at the time it is sent to Company; and (iv) information that is furnished orally by Wistar if Wistar identifies such information as "confidential" or proprietary in writing by a memorandum delivered to Company within [**] ([**]) business days after the date of disclosure.

Exclusive License Agreement Wistar/OncoCyte

- 8.1.2 "<u>Company Confidential Information</u>" means (a) the Commercialization Plan, (b) any reports prepared by Company and provided to Wistar pursuant to Sections 3.9 and 4.41, (c) any other information or material provided by Company to Wistar in tangible form that is marked as "confidential" or proprietary at the time it is delivered to Wistar; and (d) information that is furnished orally by Company if Company identifies such information as "confidential" or proprietary in writing by a memorandum delivered to Wistar within [**] ([**]) business days after the date of disclosure.
 - 8.1.3 "Confidential Information" means the Wistar Confidential Information and the Company Confidential Information, as applicable.
- 8.1.4 For the Term of this Agreement and a period of [**] ([**]) years thereafter, (i) Company shall maintain in confidence and shall not disclose to any third party any Wistar Confidential Information, and (ii) Wistar shall maintain in confidence and shall not disclose to any third party any Company Confidential Information. Each party shall take all reasonable steps to protect the Confidential Information of the other party with the same degree of care used to protect its own confidential or proprietary information. Neither party shall use the Confidential Information of the other party for any purpose other than those contemplated by this Agreement. The foregoing obligations under this Section 8.1 shall not apply to:
- (i) information that is known to the receiving party or independently developed by the receiving party prior to the time of disclosure without use of or reference to the other party's Confidential Information, in each case, to the extent evidenced by written records promptly disclosed to the furnishing party upon receipt of the Confidential Information;
 - (ii) information disclosed to the receiving party by a third party that has a right to make such disclosure;
- (iii) information that becomes patented, published or otherwise part of the public domain as a result of acts by the furnishing party or a third party obtaining such information as a matter of right; or
- (iv) information that is required to be disclosed by rule or order of the FDA or similar authority or a court of competent jurisdiction or other government authority or agency; provided that the parties shall use their best efforts to obtain confidential treatment of such information by the agency, authority, or court.
- 8.1.5 Wistar shall not be obligated to accept, and assumes no institutional liability or responsibility for, Company Confidential Information that Company furnishes to any employee of Wistar other than its business or legal officers as provided in this Agreement. If Company desires to furnish any Company Confidential Information to other employees of Wistar, Company shall so inform Wistar and Wistar shall decide whether such individual may receive some or all Company Confidential Information and, if so, whether such individual shall sign a separate confidentiality agreement to govern the use and disclosure of such information.

Exclusive License Agreement Wistar/OncoCyte

8.2 <u>Publication</u> . Company acknowledges the	at a basic objective of the research and development activities of Wistar is the generation of new
knowledge and its expeditious dissemination. To further t	that objective, Wistar retains the right, at its discretion, to demonstrate, publish or publicize a
description of the Licensed Patents and Technical Informa	ation and any results of research conducted by Wistar with or relating to the Licensed Patents or
Technical Information.	

8.3 Use of Name; Publicity.

- 8.3.1 Company shall not directly or indirectly use Wistar's name, or the name of any trustee, manager, officer, agent, employee, faculty, affiliated investigator, personnel or staff thereof, without Wistar's prior written consent, nor shall Wistar directly or indirectly use Company's name, or the name of any director, officer, agent, employee or consultant thereof, without Company's prior written consent; and
- 8.3.2 neither party shall issue any press release or other public statements related to this Agreement without the prior written consent of the other party as to each such use (which consent shall not be unreasonably withheld, conditioned or delayed).
 - 8.3.3 In each case of Sections 8.3.1 and 8.3.2 above, either party may:
- (i) make the factual statement to any third party that Company has an exclusive license from Wistar under one or more of the patents or patent applications comprising the Licensed Patents provided that no such statement shall disclose any other terms of this Agreement or any other Confidential Information of the other party; and
- (ii) provide any information that is required to be disclosed by rule or order of the FDA or similar authority or a court of competent jurisdiction or other government authority or agency (including disclosure regarding the terms of this Agreement, and filing the Agreement as an exhibit to, SEC filings), provided that the parties shall use their best efforts to obtain confidential treatment of any confidential information therein from the agency, authority or court.
- 8.3.4 Nothing set forth in this Section 8.3 shall prohibit a party to this Agreement from disclosing the existence and terms of this Agreement to its legal counsel, accountants, current or potential lenders, investment banks, or rating agencies who are subject to obligations of confidentiality.

ARTICLE 9 - TERM AND TERMINATION

9.1 Term. This Agreement shall remain in effect until the expiration of the Term unless earlier terminated as provided hereunder.

Exclusive License Agreement Wistar/OncoCyte

- 9.2 <u>Termination by Wistar</u>. Upon the occurrence of any of the events set forth below, Wistar shall have the right to terminate this Agreement by giving written notice of termination, such termination to be effective with the giving of such notice, except that in the case of Section 9.2.5 below, such termination shall occur automatically and without the necessity of notice by Wistar:
- 9.2.1 Company fails to pay any amount payable to Wistar within [**] ([**]) days after such amount becomes due; provided, however, that any failure of the Company to timely make payment of such amount shall not be deemed to be a breach of this Agreement, nor give rise to any right of Wistar to terminate this Agreement, if such failure is due to (a) Company's good faith dispute as to such amount, in whole or in part, and/or or as to whether such payment is then due under the terms of this Agreement, for which dispute the Company has provided Wistar with written notice thereof, in commercially reasonable detail, at [**] ([**]) days before the date upon which such payment is otherwise due hereunder, or (b) delay caused by or the result of any acts of God, acts of the public enemy, insurrections, riots, embargoes, labor disputes, including strikes, lockouts, job actions, or boycotts, equipment failure, fires, explosions, floods or other unforeseeable causes beyond the reasonable control and without the fault or negligence of Company, provided that Company shall give prompt notice to Wistar of such cause, and shall take whatever reasonable steps are necessary to relieve the effect of such cause as rapidly as reasonably possible and provided that the time required for delivery of such payment shall be extended for a period equal to the period of such delay, and further provided that if the failure to timely make such full payment is due to such good faith dispute by Company with respect thereto, the parties will promptly and in good faith negotiate and resolve such matter, and such payment, or portion thereof, as the parties determined by such negotiation, to be due, shall be paid by Company to Wistar within [**] ([**]) days after such mutual determination.
- 9.2.2 Company fails to pay the License Fee and the Past Patenting Costs on the Effective Date or within [**] ([**]) days after the Effective Date, as required by Sections 3.1 and 7.1.1, respectively.
- 9.2.3 material breach by Company of any covenant or agreement (other than a breach referred to in Section 9.2.1 above) or any representation or warranty contained in this Agreement that is continuing [**] ([**]) days after Wistar gives Company written notice of such breach; notwithstanding the foregoing, if Company violates the laws, regulations or other legal authority in any jurisdiction relating to the development, use, storage, or marketing of the Licensed Products in a way that Wistar deems in its reasonable judgment to constitute a public safety or health hazard, Wistar may immediately terminate the license hereunder in the applicable jurisdiction, but the remainder of the Agreement shall continue in all other jurisdictions;
 - 9.2.4 Company becomes subject to a Bankruptcy Event;
 - 9.2.5 the dissolution or cessation of operations by Company;

Exclusive License Agreement Wistar/OncoCyte

- 9.2.6 Company or any of its Affiliates, or a Sublicensee or any of its Affiliates, brings a Patent Challenge against Wistar, or assists others in bringing a Patent Challenge against Wistar (except as required under a court order or subpoena);or
- 9.2.7 Company fails to perform or fulfill its diligence obligations or any Performance Milestone in accordance with the requirements of Sections 4.1 and 4.2.
- 9.2.8 Wistar's right of termination in this Section 9.2 shall be in addition and without prejudice to, and shall not constitute a waiver of, any right of Wistar for recovery of any monies then due to it hereunder or any other right or remedy Wistar may have at law, in equity or under this Agreement.

9.3 <u>Termination by Company</u>.

- 9.3.1 Company may terminate this Agreement upon [**] ([**]) days' prior written notice to Wistar if Wistar is in material breach of this Agreement and such material breach remains uncured for [**] ([**]) days after Company gives Wistar written notice of such breach.
- 9.3.2 In addition, Company shall have the right to terminate this Agreement at any time after the second (2nd) anniversary of the Effective Date of the Agreement with or without cause upon [**] ([**]) days prior written notice to Wistar.
- 9.3.3 Company's right of termination in this Section 9.3 shall be in addition and without prejudice to, and shall not constitute a waiver of, any right or remedy Company may have at law, in equity or under this Agreement.
- 9.4 <u>Effect on Sublicenses</u>. Upon termination of this Agreement for any reason, Company shall promptly notify its Sublicensees of such termination. Upon notice by Wistar of its intent to terminate (or, if notice is not required, upon termination) of this Agreement, Company shall no longer have the authority to grant further Sublicenses. Any Sublicenses granted by Company under Section 2.4 of this Agreement shall terminate upon termination of this Agreement, unless Wistar, in its sole discretion, requests in writing that such Sublicense survive termination and remain in force and effect, in which case such Sublicense shall be assigned to Wistar.
 - 9.5 <u>Rights and Duties Upon Termination</u>. Upon termination of this Agreement for any reason:
- 9.5.1 all rights and licenses granted to Company under the terms of this Agreement shall terminate and nothing herein shall be construed to release either party from any obligation that matured prior to the effective date of such termination;
 - 9.5.2 all Confidential Information of the other party shall be promptly returned or destroyed, at the disclosing party's election;

Exclusive License Agreement Wistar/OncoCyte

- 9.5.3 Company shall cease all production and sale of Licensed Product;
- 9.5.4 final reports in accordance with Sections 3.9 and 4.4 shall be submitted to Wistar; and
- 9.5.5 all royalties and other payments, including any unreimbursed Patent Costs, accrued or due to Wistar as of the termination date shall become immediately payable.
- 9.5.6 Notwithstanding the foregoing, after the effective date of termination of this Agreement, unless for breach by Company, Company and its Sublicensees may, for a period of [**] ([**]) [**], sell all Licensed Products existing at the time of such termination, and complete Licensed Products in the process of manufacture at the time of such termination and sell the same, provided that Company shall comply with, and cause its Sublicensees to comply with, all of the terms of this Agreement, including, (a) Company shall pay to Wistar the running royalties and other payments as required hereinabove under Article 3, (b) insurance required hereunder shall be in effect, as described in Section 6.5, and (c) Company shall submit the reports required by Section 3.9 hereof.
- 9.6 <u>Disposition of Company Developments</u>. In the event this Agreement is terminated, Wistar's financial interest in and to the Licensed Patents may be harmed, due to lost patent term and other factors. Therefore, in the event of termination of this Agreement prior to expiration of the Term, Company shall:
- 9.6.1 provide Wistar with access to and deliver all documents, filings, data and other information in Company's possession directly relating to any of the Licensed Patents, Technical Information or Licensed Products, including relevant records required by regulatory authorities to be maintained with respect to Licensed Products, relevant regulatory filings, approvals, reports, records, correspondence and other regulatory materials (including any directly related to reimbursement or pricing approvals with respect to Licensed Products), and relevant documents, data and other information directly related to Clinical Trials and other studies of Licensed Products (collectively, "Documentation and Approvals");
- 9.6.2 permit Wistar and its licensees and sublicensees to utilize, reference, cross reference, incorporate in applications and filings, and otherwise have the benefit of all Documentation and Approvals; and
- 9.6.3 provide to Wistar a copy of, and grant Wistar a non-exclusive royalty-free, fully paid-up, perpetual, irrevocable, sublicensable license to, all patents and applications of Company and its Affiliates ("Company IP") that improve or are otherwise related to the Licensed Patents or that cover a Licensed Product. Wistar shall be free to use Company IP in the course of developing Licensed Products and/or otherwise exploiting the Licensed Patents, including licensing such rights to third parties.
- 9.7 <u>Provisions Surviving Termination</u>. Company's obligation to pay any amounts accrued but unpaid, and to discharge any obligations or responsibilities arising, prior to termination of this Agreement shall survive such termination. In addition, Sections 2.2.1, 3.1.1, 4.5.1, 7.2, 9.4, 9.5, 9.6, 9.7, 9.8, 10.3, 10.4 and 10.8 and Articles 6 and 8, the defined terms and provisions used or referenced therein, and any other provisions required to interpret the rights and obligations of the parties arising prior to the termination date shall survive expiration or termination of this Agreement.

Exclusive License Agreement Wistar/OncoCyte

Right to Payment Accrues During Term of Agreement. Whenever a payment to Wistar with respect to sales of any Licensed Product is provided for in this Agreement, the right of Wistar to such payment shall accrue at the time such product is manufactured or produced or service is provided during the Term of this Agreement, provided that such payment shall not be due until the time otherwise provided therefor in this Agreement. Any inventory or stocks of Licensed Product existing prior to the expiration of this Agreement but sold after expiration of this Agreement shall generate payment to Wistar in accordance with the applicable percentage or other method for determining the amount of such payment provided in this Agreement, and provided that (i) Company shall promptly remit the required payment to Wistar after the post-expiration sale of such Licensed Product, and (ii) after expiration of this Agreement, Company may destroy any inventory or stocks remaining at such expiration which are not needed to fill orders for Licensed Products that are open and unfilled by Company as of such expiration and (iii) Company may at its election continue to sell, after such expiration, inventory or stocks of Licensed Product existing prior to such expiration, until the earlier of (a) all such inventory or stocks have been sold or any then remaining inventory or stocks have been destroyed, or (b) [**] ([**]) [**] from the termination date of this Agreement.

ARTICLE 10 - ADDITIONAL PROVISIONS

10.1 Assignment.

- 10.1.1 This Agreement shall be binding upon and shall inure to the benefit of the parties and their respective permitted assigns and successors in interest. Except as expressly permitted in this Agreement, Company shall not assign, delegate or subcontract any of its rights or obligations under this Agreement without the prior written consent of Wistar.
- 10.1.2 No such consent shall be required to assign this Agreement to a successor in connection with a merger or consolidation of Company, or to the purchaser of all or substantially all the assets of Company, provided that: (i) Company is not in breach of this Agreement; (ii) such successor or purchaser shall agree in writing to be bound by the terms and conditions hereof prior to such assignment; (iii) Company shall provide Wistar with evidence to demonstrate that such successor or purchaser has or is likely to acquire, in a reasonable period of time, capital and personnel resources sufficient to fulfill the obligations it is assuming hereunder; and (iv) Company shall notify Wistar in writing of any assignment and provide a copy of all assignment documents (pursuant to which such transferee shall have agreed in writing to be bound by the terms and conditions of this Agreement) to Wistar within thirty (30) days of assignment.

Wistar Reference No. LIC15-35

Exclusive License Agreement Wistar/OncoCyte

- 10.1.3 Failure of an assignee to agree to be bound by the terms hereof or failure of Company to notify Wistar and provide copies of assignment documentation shall be grounds for termination of this Agreement for default. Any attempted assignment in contravention of this Section 10.1.3 shall be null and void.
- 10.2. <u>No Waiver</u>. A waiver by either party of a breach or violation of any provision of this Agreement shall not constitute or be construed as a waiver of any subsequent breach or violation of that provision or as a waiver of any breach or violation of any other provision of this Agreement.
- 10.3 <u>Independent Contractor</u>. Nothing herein shall be deemed to establish a relationship of principal and agent between Wistar and Company, nor any of their Sublicensees, agents or employees for any purpose whatsoever. This Agreement shall not be construed as constituting Wistar and Company as partners, or as creating any other form of legal association or arrangement which could impose liability upon one party for the act or failure to act of the other party.
- Notices. Any notice and all invoices given under this Agreement shall be in writing and shall be deemed delivered when sent by prepaid, express, first class, certified or registered mail, or by overnight courier, with confirmed receipt, addressed to the parties as follows, or by email where shown below with regards to invoices to Company (or at such other addresses as the parties may notify each other in writing pursuant to this Section 10.4):

If to Wistar:

The Wistar Institute
3601 Spruce Street
Philadelphia, PA 19104
Attn: Director, Business Development
With Copy to: Vice President, Legal and External Affairs

If to Company:

OncoCyte Corporation 1301 Harbor Bay Parkway Alameda, CA 94502 Attn: Chief Executive Officer

Invoices to be sent to [**]

10.5 <u>Entire Agreement</u>. This Agreement embodies the entire understanding between the parties relating to the subject matter hereof and supersedes all prior understandings and agreements, whether written or oral. This Agreement may not be varied except by a written document signed by duly authorized representatives of both parties.

Exclusive License Agreement Wistar/OncoCyte

- 10.6 <u>Severability</u>. Any of the provisions of this Agreement which are determined to be invalid or unenforceable in any jurisdiction shall be ineffective to the extent of such invalidity or unenforceability in such jurisdiction, without rendering invalid or unenforceable the remaining provisions hereof or affecting the validity or unenforceability of any of the terms of this Agreement in any other jurisdiction.
- 10.7 <u>Headings; Interpretation</u>. Any article and section headings and captions used in this Agreement are for convenience of reference only and shall not affect its construction or interpretation. The words "include" or "including" shall be construed as incorporating, also, "but not limited to" or "without limitation." The parties acknowledge that each party has read and negotiated the language used in this Agreement. Because all parties participated in negotiating and drafting this Agreement, no rule of construction shall apply to this Agreement which construes ambiguous language in favor of or against any party by reason of that party's role in drafting this Agreement.
- 10.8 <u>No Third Party Benefits</u>. Nothing in this Agreement, express or implied, is intended to confer on any Person other than the parties hereto or their permitted assigns, any benefits, rights or remedies.
 - 10.9 <u>Disputes; Governing Law; Jurisdiction</u>.
- 10.9.1 In the case of any dispute, claim, question or disagreement arising out of or relating to this Agreement, or the parties' activities hereunder, including any question regarding the existence, validity or termination of this Agreement, the parties shall use all reasonable efforts to settle such dispute, claim, question or disagreement by amicable agreement, including by escalation to the President and Chief Executive Officer of Wistar and Chief Executive Officer of Company, if necessary, prior to commencement of litigation.
- 10.9.2 This Agreement shall be construed, governed, interpreted and applied in accordance with the laws of the Commonwealth of Pennsylvania, without giving effect to conflict of law principles.
- 10.9.3 Each party (i) irrevocably submits to the exclusive jurisdiction of the United States District Court for the Eastern District Pennsylvania or a local court sitting in the city of Philadelphia, Pennsylvania (collectively, the "Courts") for purposes of any action, suit or other proceeding relating to or arising out of this Agreement and (ii) agrees not to raise any objection at any time to the laying or maintaining of the venue of any such action, suit or proceeding in any of the Courts, irrevocably waives any claim that such action, suit or other proceeding has been brought in an inconvenient forum and further irrevocably waives the right to object, with respect to such action, suit or other proceeding, that such Court does not have any jurisdiction over such party.
- 10.10 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original as against the party whose signature appears thereon, but all of which taken together shall constitute but one and the same instrument.

Exclusive License Agreement Wistar/OncoCyte

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the Effective Date.

THE WISTAR INSTITUTE OF ANATOMY AND BIOLOGY

/s/ Heather A. Steinman

Heather A. Steinman, Ph.D., M.B.A. Vice President, Business Development Executive Director, Technology Transfer

Date: January 22, 2016

By:

ONCOCYTE CORPORATION

By: /s/ William Annett
Name: William Annett

Title: CEO

Date: January 22, 2016

Exclusive License Agreement Wistar/OncoCyte

Page 30 of 32

EXHIBIT A

Licensed Patents

Tech ID	Serial Number	Filing Date	Title	Inventors
15-12	62/163,766	May 19, 2015	Methods and Compositions for Diagnosing or Detecting Lung Cancers	Drs. Louise Showe, Michael K. Showe, Andrei V. Kossenkov

Exclusive License Agreement Wistar/OncoCyte

EXHIBIT B

Technical Information

The [**] information provided by Wistar to Company under the memorandum of understanding entered into by and between the parties on December 29, 2015

Exclusive License Agreement Wistar/OncoCyte

CERTIFICATIONS

I, Michael D. West, certify that:

- 1. I have reviewed this annual report on Form 10-K/A of BioTime, Inc.
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

Date: May 24, 2016

/s/ Michael D. West

Michael D. West

Co-Chief Executive Officer

CERTIFICATIONS

I, Aditya Mohanty, certify that:

- I have reviewed this annual report on Form 10-K/A of BioTime, Inc.
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

Date: May 24, 2016

/s/ Aditya Mohanty Aditya Mohanty

Co-Chief Executive Officer

CERTIFICATIONS

I, Russell Skibsted, certify that:

- 1. I have reviewed this annual report on Form 10-K/A of BioTime, Inc.
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

Date: May 24, 2016

/s/ Russell Skibsted

Russell Skibsted Chief Financial Officer